**Author's response to reviews**

**Title:** Development and initial testing of a Gestational Diabetes Mellitus (GDM) information website for multi-ethnic and low socio-economic women

**Authors:**

Mary C Carolan-Olah (mary.carolan@vu.edu.au)
Cheryl Steele (Cheryl.Steele@wh.org.au)
Heather Margetts (Heather.margetts@wh.org.au)
Gillian Krenzin (Gillian.Krenzin@nh.org.au)

**Version:** 2  **Date:** 24 October 2014

**Author's response to reviews:**

Response to suggested revisions

I have inserted the line numbers as requested and incorporated the tables into the manuscript.